Author's response to reviews

Title: The use of instrumented gait analysis for individually tailored interdisciplinary interventions in children with cerebral palsy: A randomised controlled trial protocol

Authors:

Helle M Rasmussen (helrasmussen@health.sdu.dk)
Niels W Pedersen (niels.w.pedersen@rsyd.dk)
Søren Overgaard (soeren.overgaard@rsyd.dk)
Lars K Hansen (lars.kjaersgaard.hansen@rsyd.dk)
Ulrike Dunkhase-Heinl (ulrike.dunkhase-heinl@rsyd.dk)
Yanko Petkov (yanko.petkov@rsyd.dk)
Vilhelm Engell (vilhelm.engell@rsyd.dk)
Richard Baker (r.j.baker@salford.ac.uk)
Anders Holsgaard-Larsen (ahlarsen@health.sdu.dk)

Version: 5
Date: 24 November 2015

Author's response to reviews: see over
The use of Instrumented gait analysis for individually tailored interdisciplinary interventions in children with cerebral palsy: A randomised controlled trial protocol

To the Editor,

We have placed all figure legends at the end of the main manuscript, after References and removed them from the figure files.

On behalf of all Authors

Helle Mätzke Rasmussen

Physiotherapist, Ph.D.-Student

Department of Orthopaedic Surgery and Traumatology, Odense University Hospital, Denmark and Institute of Clinical Research, University of Southern Denmark, Odense, Denmark.